Author's response to reviews

Title: Epidemiology of cardiovascular risk factors in Greece; aims, design and baseline characteristics of the ATTICA study

Authors:

Dr Christos Pitsavos (d.b.panagiotakos@usa.net)
Dr Demosthenes B Panagiotakos (d.b.panagiotakos@usa.net)
Dr Christina Chrysohoou (d.b.panagiotakos@usa.net)
Dr John Skoumas (d.b.panagiotakos@usa.net)
Dr Christodoulos Stefanadis (d.b.panagiotakos@usa.net)

Version: 3 Date: 25 Sep 2003

Dear Editor

Thank you very much for your letter. We are greatfull for your decision to publib our paper in your journal.
We have made all the requested changes.

Yours truly,

Demosthenes B Panagiotakos

Replies to the Editor's comments

Formatting changes requested
----------------------------

COM 1: Title page:Please remove the first comma from the title, changing it to:
Epidemiology of cardiovascular risk factors in Greece; aims, design and baseline characteristics of the ATTICA study
DONE

COM 2: Affiliation- It is not clear if all authors are at the same institution.
Please confirm, and if this is not the case please provide each affiliation written in full and in the following format: Department, Institute, City, Country. Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers
ALL AUTHORS ARE FROM THE SAME INSTITUTION, WE LINKED EACH AUTHOR TO THE AFFILIATION

COM 3: Please remove the supplementary file
WE REMOVED THE SUPPLEMENTARY FILES

COM 4: The borders of Tables included as part of the main manuscript must be visible black lines. This will allow us to replicate your table layout more accurately in the final version. I would like to reassure you that the final style of the tables will be much the same as their present style.
WE HAVE CHANGED THE STYLE OF THE TABLES ACCORDING TO YOUR SUGGESTION
COM 5: Please do check your manuscript carefully for any typos. We have spotted a number of small errors throughout the manuscript. I have made a few changes to the background of your abstract as an example:
WE ASKED AN ENGLISH SPEAKER FRIEND TO REVISE OUR MS....WE MADE AN EFFORT TO CORRECT ALL GRAMMAR AND SPELLING ERRORS..THANK YOU VERY MUCH FOR YOUR HELP...

COM 6: Please be consistent in your use of spelt out numbers versus numeric characters. Please use one or the other, irrespective of where in the sentence it comes.
WE CHANGE ALL NUMBERS INTO NUMERIC CHARACTERS...

COM 7: Introduction - Please change this to 'Background'
All section headings- Please use English style capitalization. Only the first word and any proper nouns or acronyms should be capitalised. Please do not use block capitals
WE HAVE CHANGED ALL SECTION HEADINGS ACCORDING TO YOUR REQUEST...

THANK YOU ONCE AGAIN FOR YOUR TIME....